Massachusetts Institute of Technology Handout 25 6.857: Network and Computer Security November 3, 1998 Professor Ronald L. Rivest

Readings for Lecture 15

Attached are the following articles:

• Quantum Cryptography, pages 50–57. [1]

• Underwater quantum coding, page 449. [2]

References

[1] Charles H. Bennett, Gilles Brassard, and Artur K. Ekert. Quantum cryptography. Scientific American, 267(4):50–57, 1992.

[2] A. Muller, H. Zbinden, and N. Gisin. Underwater quantum coding. Nature, 378:449, 1995.